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Abstract 

Manufacturing biopharmaceuticals is a complex process. Monitoring and controlling of process 

variables is necessary to ensure consistent quality of the product. However, in-process quality 

control is traditionally limited to off line analysis. This results in delay in taking critical 

decisions, batch failures, and at the times loss of valuable product. In view of this, regulatory 

authorities have proposed process analytical technology (PAT) guidance for the biotech 

industries with an aim of achieving consistent product quality. PAT offers an approach to 

facilitate application of process knowledge for improvement of control, delivering consistent 

product quality, and delivering superior operational efficiency. 

Implementation of PAT requires establishment of process models, mathematical relationships 

between critical quality attributes, critical process parameters and feed material attributes. 

While mechanistic models are desirable, most biotech unit operations are too complex to create 

mechanistic models that are simple enough for use in commercial manufacturing. Use of 

empirical models or hybrid models is more of a norm in industry. The two major platforms 

used in biopharmaceutical manufacturing are the bacterial Escherichia coli system, and the 

mammalian Chinese Hamster Ovary system. In E. Coli based process, refolding is one of the 

important unit operations which decides the yield, impurity profile of the products. 

Protein refolding is known to be significantly impacted by critical process parameters and feed 

material attributes including composition and pH of the solubilisation and refolding buffers. 

Hence, to achieve robust process control and product quality, these attributes and parameters 

need to be monitored. A PAT tool has been developed using online oxidation reduction 

potential, pH, and temperature probes to create a fingerprint of a typical refolding batch within 

the pre-defined design and operating spaces. A database of successful runs about a set-point is 

then created and used to develop statistical process control charts to monitor the evolution of 

the batch with respect to the addition of each subsequent component. These charts have been 

used for real-time monitoring of process deviations. This prevents wastage of valuable 

inclusion body or solubilized inclusion bodies material and saves time by allowing early 

detection of the otherwise unsalvageable batches. The tool also allows a history of batch 

information to be saved in the form of easily interpretable profiles for future reference. The 

probes are robust, cost-effective, and can be easily integrated with data collection systems. 

Using these three online indicators in real time has been demonstrated to ensure that the process 

is being operated within the established design space, thereby potentially eliminating or at least 
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minimizing the need for testing of the product critical quality attributes using offline methods 

such as reverse phase-high performance liquid chromatography.  

Many times biotherapeutic faces challenges such as lower half life and stability inside the body. 

PEGylation of the product is a well-known method to improve the pharmacological properties 

and stability of biotherapeutic proteins. Proper control of the PEGylation reaction step is critical 

as the molecule’s size and nature changes significantly during the reaction, as does the impurity 

profile of the process stream. A near infrared spectroscopy-based PAT approach has been 

developed for monitoring and control of the PEGylation reaction of G-CSF. NIRS spectra have 

been used to build calibrated and validated PLS models for the three species involved in the 

reaction, namely G-CSF and its monoPEGylated and diPEGylated variants. A NIRS immersion 

probe was deployed in the reaction vessel to provide real time information about the progress 

of the reaction. A pH probe integrated with a peristaltic pump facilitated automated quenching 

of the reaction at the deserved time. The NIRS spectra have also been used to build a Batch 

Evolution Model for the reaction from end-to-end, including the addition of the reactants to the 

reaction vessel, the progress of the reaction for 70 minutes, and the final quenching with Tris 

base. Comparing real-time data with the expected trajectory of statistical process control charts 

from the Batch Evolution Model allowed for the deviations to be identified as and when they 

occur, and aid in root cause analysis. 

Snakebite has been a neglected disease for decades until the World Health Organization 

included it in the list of tropical neglected diseases in 2009. A process for manufacturing the 

recombinant Lethal Toxin Neutralizing factor (rLTNF) has been developed with overall 

recovery of 78–80%. A manufacturing plant capable of producing 10 Kg of rLTNF per batch 

was simulated and techno economic analysis performed. The production cost of rLTNF was 

estimated to be $5–6 per g (or $0.09 per 15 mg dose), significantly lower in cost to that of 

currently available antivenom in the market (approximately $6 per dose).Further, a novel three-

step process for manufacturing of rLTNF monomer to reduce the manufacturing cost by 30%. 

The tolerance of alpha chymotrypsin enzyme to denaturing conditions was used to develop a 

simultaneous solubilization and cleavage reaction in single step, in which the denaturant urea 

not only aids inclusion body solubilization but also opens the peptide chains for efficient 

cleavage. The cleavage reaction was identified to be critical as it determines product yield and 

purity, and thus all further downstream processes and was optimized using Design of 

Experiments techniques. The proposed process offers reduced manufacturing cost by up to 30% 

from the previously published process and can be easily adapted to any peptide manufacturing 

process. Finally, a PAT implementation for monitoring and control of the rLTNF enzymatic 
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cleavage reaction was developed using ATR-FTIR spectroscopy which provides distinctive 

spectra for rLTNF concatemer and urea concentration in the reaction mixture. PLS calibration 

models were built to correlate the at-line spectra with concentration of rLTNF concatemer, 

allowing the progress of the reaction and percentage conversion to be tracked in real time. PAT 

implementation ensured that the product formed meets critical quality attribute targets of purity, 

yield and impurity profile. The tool also enabled identification of deviations, such as deviations 

in the reaction buffer, concentration of concatemer in the IBs, enzyme loading relative to 

protein concentration, and reaction time with late quenching, enabling robust monitoring and 

control. 

In protein therapeutics, it is critical that the excipient concentrations in the drug product are 

maintained within the prescribed specifications. This task is quite straightforward when the 

protein concentration in drug product is low. However, as the protein concentrations rise in the 

drug product, especially in the case of subcutaneous formulations, maintaining precise levels 

of excipients becomes challenging. At higher concentration (in excess of 100 mg/ml), protein 

molecules tend to attract/repel the charged excipients when processed through tangential flow 

filtration membrane, resulting in a significant drift of excipient concentration in the final 

formulated drug substance. This in turn alters the pH, conductivity, and osmolality of the 

solution. Limited literature exists on methods for analysis of excipients in protein drug products 

and this is so due to incompatibility of protein with organic solvents and wide diversity with 

respect to the nature of excipients used in protein therapeutic formulations. Two different 

chromatographic modes, ion exclusion chromatography and reversed phase chromatography, 

were deployed to analyse a total of seventeen excipients that are widely present in protein 

formulations ranging from peptides to monoclonal antibodies. Both the methods were validated 

as per ICH guidelines. The present methods did not require sample pre-treatment. The 

application of the methods has been demonstrated by analysing excipients concentration in 

various innovator products and their biosimilars. A mechanistic model incorporating coupling 

of the charge fluxed through the membrane was developed to explain the drift in the excipient 

concentration observed in DF and UF operations in the processing of mAbs. It was found that 

the drift of the excipients in both the DF and UF operations is most strongly influenced by their 

diffusivity, followed by the charge on the mAb and its concentration. The initial/ exchange 

buffer concentration also had a significant influence on the drift of the excipients. The model 

was validated with experimental results, and the match between the experimental and the 

model-simulated values were within 8%. It has been demonstrated that the model can be used 

to predict the initial/ exchange buffer concentration for a target final concentration after the 
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UF/DF operation. The effect of system parameters on the drift of the excipients was determined 

from the model, and the diffusivities of the excipients and mAb charge were found to have the 

strongest influence. 

We believe that the studies presented here as a part of this thesis will promote development of 

highly efficient PAT tools for manufacturing of biotherapeutics. 

 

  



सार 

विविर्माण बमयोफमर्मास्यवुिकल्स एक जविल प्रवियम ह।ै उत्पमद की विरंतर गुणित्तम सुविवित करि ेके वलए प्रवियम चर की विगरमिी और वियंत्रण 

आिश्यक ह।ै हमलमंवक, इि-प्रोसेस गुणित्तम वियंत्रण पमरंपररक रूप से ऑफ लमइि विशे्लषण तक सीवर्त ह।ै यह र्हत्िपणूा विणायों, बैच विफलतमओ,ं 

और र्लू्यिमि उत्पमद के िकुसमि के सर्य र्ें देरी को रोकतम ह।ै इसे देखते हुए, वियमर्क अविकमररयों िे लगमतमर उत्पमद गुणित्तम प्रमप्त करि ेके 

उद्देश्य से बमयोिेक उद्योगों के वलए प्रवियम विशे्लषणमत्र्क प्रौद्योवगकी (प्याट) र्मगादर्ाि कम प्रस्तमि वकयम ह।ै पीएिी वियंत्रण के सिुमर, विरंतर उत्पमद 

की गुणित्तम प्रदमि करिे और बेहतर पररचमलि दक्षतम प्रदमि करि ेके वलए प्रवियम ज्ञमि के अिपु्रयोग की सुवििम के वलए एक दृविकोण प्रदमि करतम 

ह।ै 

पीएिी के कमयमान्ियि के वलए प्रवियम र्ॉडल की स्थमपिम, र्हत्िपणूा गुणित्तम विरे्षतमओ,ं र्हत्िपणूा प्रवियम र्मपदंडों और फीड समर्ग्री विरे्षतमओ ं

के बीच गवणतीय संबंि की आिश्यकतम होती ह।ै जबवक रै्केविवस्िक र्ॉडल िमंछिीय हैं, अविकमंर् बमयोिेक यवूिि संचमलि यमंवत्रकी र्ॉडल बिमिे 

के वलए बहुत जविल हैं जो िमवणवययक विर्माण र्ें उपयोग के वलए पयमाप्त सरल हैं। अिभुिजन्य र्ॉडल यम हमइविड र्ॉडल कम उपयोग उद्योग र्ें एक 

आदर्ा ह।ै बमयोफर्मावसविकल रै्न्यफैुक्चररंग र्ें इस्तेर्मल वकए जमि ेिमले दो प्रर्ुख प्लेिफॉर्ा हैं बैक्िीररयल एस्चेररवचयम कोलमई वसस्िर् और स्तििमरी 

चीिी हम्सिर ओिेररयि वसस्िर्। ई। कोलमई आिमररत प्रवियम र्ें, रेफोल्डिंग उि र्हत्िपणूा इकमइयों के संचमलि र्ें से एक ह ैजो उत्पमदों की उपज, 

अर्दु्धतम प्रोफमइल कम फैसलम करतम ह।ै 

प्रोिीि ररफॉवल्डंग को र्हत्िपणूा प्रवियम र्मपदंडों और फीड समर्ग्री विरे्षतमओ ंसे प्रभमवित होिे के वलए जमिम जमतम है, वजसर्ें सॉल्यवूबलेर्ि और 

पीएच की संरचिम और बफर को र्मवर्ल वकयम जमतम ह।ै इसवलए, र्जबूत प्रवियम वियंत्रण और उत्पमद की गुणित्तम प्रमप्त करि े के वलए, इि 

विरे्षतमओ ंऔर र्मपदंडों पर िजर रखि ेकी आिश्यकतम ह।ै पिूा-वििमाररत वडजमइि और ऑपरेविंग ररक्त स्थमि के भीतर एक विवर्ि रीफॉवल्डंग 

बैच कम एक वफंगरवपं्रि बिमिे के वलए ऑिलमइि ऑक्सीकरण कर्ी क्षर्तम, पीएच और तमपर्मि जमंच कम उपयोग करके एक पैि िूल विकवसत 

वकयम गयम ह।ै सेि-पॉइिं के बमरे र्ें सफल रि कम एक डेिमबेस तब बिमयम जमतम ह ैऔर प्रत्येक बमद के घिक के अलमिम के समथ बैच के विकमस 

की विगरमिी के वलए समंवययकीय प्रवियम वियंत्रण चमिा विकवसत करि ेके वलए उपयोग वकयम जमतम ह।ै इि चमिों कम उपयोग प्रवियम विचलि की 

िमस्तविक सर्य की विगरमिी के वलए वकयम गयम ह।ै यह र्लू्यिमि सर्मिेर्ि के अपव्यय को रोकतम ह ैयम सर्मिेर्ी विकमय समर्ग्री को घवुलत करतम 

ह ैऔर अन्यथम अप्रमप्य बैचों कम र्ीघ्र पतम लगमिे की अिरु्वत देकर सर्य बचमतम ह।ै उपकरण भविष्य के संदभा के वलए आसमिी से व्यमययम करिे 

योग्य प्रोफमइल के रूप र्ें बैच जमिकमरी के इवतहमस को सहजेि ेकी अिरु्वत देतम ह।ै जमंच र्जबूत, लमगत प्रभमिी ह,ै और आसमिी से डेिम संग्रह 

प्रणमवलयों के समथ एकीकृत वकयम जम सकतम ह।ै िमस्तविक सर्य र्ें इि तीि ऑिलमइि संकेतकों कम उपयोग करके यह सवुिवित करि ेके वलए 

प्रदर्ाि वकयम गयम ह ैवक यह प्रवियम स्थमवपत वडजमइि स्थमि के भीतर संचमवलत हो रही ह,ै वजससे ऑफलमइि चरण कम उपयोग करते हुए उत्पमद 

र्हत्िपणूा गुणित्तम के परीक्षण की आिश्यकतम को कर् से कर् यम कर् से कर् वकयम जम सके, जैस ेवक ररिसा चरण- उच्च उत्पमदि द्रव्य िणालेखि। 

कई बमर जैि-वचवकत्सीय चिुौवतयों कम समर्िम करिम पड़तम ह,ै जैस ेविचल ेआि ेजीिि और र्रीर के अंदर वस्थरतम। औषिीय गुणों को बेहतर 

बिमिे के वलए और जैि-वचवकत्सीय प्रोिीि की वस्थरतम को बेहतर बिमिे के वलए उत्पमद कम पेगीलेर्ि एक प्रवसद्ध विवि ह।ै अण ुप्रवतवियम के चरण 

कम उवचत वियंत्रण र्हत्िपणूा ह ैक्योंवक प्रवतवियम के दौरमि अण ुकम आकमर और प्रकृवत र्ें र्हत्िपणूा पररिताि होतम ह,ै जैसम वक प्रवियम स्रीर् की 

अर्दु्धतम प्रोफमइल करतम ह।ै जी-सीएसएफ के पेगीलेर्ि प्रवतवियम की विगरमिी और वियंत्रण के वलए एक विकि अिरक्त स्पेक्रोस्कोपी-आिमररत 

पैि दृविकोण विकवसत वकयम गयम ह।ै एनआईआरयस स्पेक्रम कम उपयोग प्रवतवियम र्ें र्मवर्ल तीि प्रजमवतयों के वलए कैवलिेिेड और िैवलडेिेड 

पीएलएस र्ॉडल के विर्माण के वलए वकयम गयम ह,ै अथमात् जी-सीएसएफ और इसके र्ोिोपेगलेिेड और डमयपेगलेिेड िेररएंि। प्रवतवियम की प्रगवत 



के बमरे र्ें िमस्तविक सर्य की जमिकमरी प्रदमि करि ेके वलए प्रवतवियम पोत र्ें एक एिआईआरएस विसजाि जमंच तैिमत की गई थी। एक पीएच 

जमंच एक िवर्क ितृ्तों र्ें वसकुड़िेिमलम पंप के समथ एकीकृत, योग्य सर्य पर प्रवतवियम की स्िचमवलत र्र्ि की सवुििम। येनआईरयस स्पेक्रम कम 

उपयोग एंड-िू-एंड से प्रवतवियम के वलए एक बैच इिोल्यरू्ि र्ॉडल बिमिे के वलए भी वकयम गयम ह,ै वजसर्ें ररएक्िर पोत पर ररएक्िसा को जोड़िम, 

70 वर्िि के वलए प्रवतवियम की प्रगवत और वरस बेस के समथ अंवतर् र्र्ि र्मवर्ल ह।ै । बैच विकमस र्ॉडल से समंवययकीय प्रवियम वियंत्रण चमिा 

की अपेवक्षत प्रके्षपिि के समथ िमस्तविक सर्य के आंकड़ों की तुलिम विचलि के वलए अिरु्वत दी जमती ह ैजब िे होते हैं और र्लू कमरण विशे्लषण 

र्ें सहमयतम करते हैं। 

सपादंर् दर्कों से एक उपेवक्षत बीर्मरी ह,ै जब तक वक २००९ र्ें विश्व स्िमस््य संगठि िे इसे उष्णकविबंिीय उपेवक्षत बीर्मररयों की सचूी र्ें र्मवर्ल 

िहीं वकयम थम। पिुः संयोजक घमतक िॉवक्सि न्यरूमलमइवजंग फैक्िर (आरएललटयेनऍफ़) के विर्माण की एक प्रवियम को ७०-८० % की सर्ग्र िसलूी 

के समथ विकवसत वकयम गयम ह।ै प्रवत बैच १० टन्न आरएललटयेनऍफ़ के उत्पमदि र्ें सक्षर् एक विविर्माण संयंत्र कम अिकुरण वकयम गयम और 

तकिीकी आवथाक विशे्लषण वकयम गयम। आरएलिीएिएफ की उत्पमदि लमगत $ ५-६ प्रवत ग्रमर् (यम $ ०.०९ प्रवत १५ वर्लीग्रमर् खरुमक) होि ेकम 

अिरु्मि थम, बमजमर र्ें ितार्मि र्ें उपलब्ि एंिीिेिर् (लगभग $ ६ प्रवत खरुमक) की लमगत से कमफी कर् ह।ै एक उपन्यमस तीि-चरण विविर्माण 

लमगत को 30% तक कर् करिे के वलए आरएललटयेनऍफ़ र्ोिोर्र के विर्माण की प्रवियम। र्तों को विरूवपत करिे के वलए अल्फम कमइर्ोवरवप्सि 

एंजमइर् की सवहष्णतुम कम उपयोग एकल चरण र्ें एक समथ घलुिर्ीलतम और दरमर की प्रवतवियम को विकवसत करिे के वलए वकयम गयम थम, वजसर्ें 

वडिैब्यिूेंि यरूरयम ि केिल र्रीर को जोड़ि ेर्ें सहमयक होतम ह,ै बवल्क कुर्ल दरमर के वलए पेप्िमइड श्ृंखलम भी खोलतम ह।ै दरमर की प्रवतवियम को 

र्हत्िपणूा र्मिम गयम क्योंवक यह उत्पमद की उपज और र्दु्धतम को वििमाररत करतम ह,ै और इस प्रकमर आगे की सभी डमउिस्रीर् प्रवियमओ ंऔर 

वडजमइि ऑफ एक्सपेररर्ेंि तकिीकों कम उपयोग करके अिकूुवलत वकयम गयम थम। प्रस्तमवित प्रवियम पहले से प्रकमवर्त प्रवियम से विविर्माण लमगत 

को ३०% तक कर् कर देती ह ैऔर इसे आसमिी से वकसी भी पेप्िमइड विर्माण प्रवियम के वलए अिकूुवलत वकयम जम सकतम ह।ै अंत र्ें, आरिीिीएिएफ 

एंजमइरै्विक क्लीिेज ररएक्र्ि की विगरमिी और वियंत्रण के वलए एक पैि कमयमान्ियि एिीआर-एफिीआईआर स्पेक्रोस्कोपी कम उपयोग करके 

विकवसत वकयम गयम थम जो प्रवतवियम वर्श्ण र्ें आरएलएििीएि एफिीएिर और यरूरयम एकमग्रतम के वलए विवर्ि स्पेक्रम प्रदमि करतम ह।ै लियलयस 

कैवलिेर्ि र्ॉडल को आरएललटयेनऍफ़ कंसरेिर की एकमग्रतम के समथ ऑि-लमइि स्पेक्रम को सहसंबंवित करिे के वलए बिमयम गयम थम, वजससे 

प्रवतवियम और प्रवतर्त रूपमंतरण की प्रगवत को िमस्तविक सर्य र्ें रैक वकयम जम सके। पीएिी कमयमान्ियि िे सवुिवित वकयम वक गवठत उत्पमद 

र्दु्धतम, उपज और अर्ुद्धतम प्रोफमइल के र्हत्िपणूा गुणित्तम विर्ेषतम लक्ष्यों को परूम करतम ह।ै उपकरण िे विचलि की पहचमि को भी सक्षर् वकयम, 

जैस ेवक प्रवतवियम बफर र्ें विचलि, आईबी र्ें कंसरेिर की एकमग्रतम, प्रोिीि एकमग्रतम के समपेक्ष एंजमइर् लोड करिम, और देर से र्र्ि के समथ 

प्रवतवियम सर्य, र्जबूत विगरमिी और वियंत्रण को सक्षर् करिम। 

प्रोिीि वचवकत्सम विज्ञमि र्ें, यह र्हत्िपणूा ह ैवक दिम उत्पमद र्ें वििमाररत समंद्रतम वििमाररत विविदेर्ों के भीतर बिमए रखी जमती ह।ै ड्रग उत्पमद र्ें 

प्रोिीि की र्मत्रम कर् होि ेपर यह कमया कमफी सीिम ह।ै हमलमंवक, दिम उत्पमद र्ें प्रोिीि समंद्रतम र्ें िवृद्ध के रूप र्ें, विर्ेष रूप से चर्डे़ के िीच ेयोगों 

के र्मर्ले र्ें, एक्ससलिएिंट्स के सिीक स्तर को बिमए रखि ेचिुौतीपणूा ह।ै उच्च समंद्रतम र्ें (१०० लिली ग्राि / लिली लीटर से अविक), प्रोिीि अणु 

स्पर्ारेखम प्रिमह विस्पंदि विल्ली के र्मध्यर् से संसमवित होि ेपर आिेवर्त िमहकों को आकवषात / प्रवतकवषात करते हैं, वजसके पररणमर्स्िरूप अंवतर् 

तैयमर दिम पदमथा र्ें प्रिमहक एकमग्रतम कम एक र्हत्िपणूा बहमि होतम ह।ै यह बदल ेर्ें पीएच, चमलकतम, और सर्मिमि की परमसरणतम को बदल देतम 

ह।ै प्रोिीि दिम उत्पमदों र्ें एक्ससलिएिंट्स के विशे्लषण के वलए सीवर्त समवहत्य र्ौजदू ह ैऔर ऐसम कमबाविक सॉल्िैंि्स के समथ प्रोिीि की असंगवत 

और प्रोिीि वचवकत्सीय योगों र्ें उपयोग वकए गए एक्ससलिएिंट्स की प्रकृवत के संबंि र्ें व्यमपक विविितम के कमरण होतम ह।ै दो अलग-अलग 



िोरै्िोग्रमवफक र्ोड, आयि अपिजाि िोरै्िोग्रमफी और उल्िे चरण िोरै्िोग्रमफी, कुल सत्रह एक्ससलिएिंट्स कम विशे्लषण करि ेके वलए तैिमत वकए 

गए थे जो वक व्यमपक रूप से पेप्िमइड्स से र्ोिोक्लोिल एंिीबॉडी तक के प्रोिीि योगों र्ें र्ौजदू हैं। दोिों विवियों को आईसीएच वदर्मविदेर्ों के 

अिसुमर र्मन्य वकयम गयम थम। ितार्मि विवियों र्ें िर्िूम पिूा उपचमर की आिश्यकतम िहीं थी। विवभन्ि इिोिेिर उत्पमदों और उिके बमयोवसवर्लर र्ें 

एक्ससलिएिंट्स एकमग्रतम कम विशे्लषण करके विवियों के आिेदि कम प्रदर्ाि वकयम गयम ह।ै विल्ली के र्मध्यर् से प्रिमवहत आिेर् के यगु्र्ि को 

सवम्र्वलत करि ेिमलम एक यंत्रित र्ॉडल विकवसत वकयम गयम थम तमवक ियाबस के प्रसंस्करण र्ें लडएफ और यूएफ वियमओ ंर्ें र्िमयम जमिे िमले 

उत्कृि समंद्रतम र्ें बहमि को सर्िमयम जम सके। यह पमयम गयम वक लडएफ और यूएफ संचमलि दोिों र्ें एक्सवफवलएंि्स कम बहमि उिके वडफ्यूवसिी से 

सबस ेअविक प्रभमवित होतम ह,ै इसके बमद ियाबस और उसके समंद्रतम पर चमजा होतम ह।ै प्रमरंवभक / विविर्य बफर एकमग्रतम कम भी एक्ससलिएिंट्स 

के बहमि पर एक र्हत्िपणूा प्रभमि थम। र्ॉडल को प्रयोगमत्र्क पररणमर्ों के समथ र्मन्य वकयम गयम थम, और प्रयोगमत्र्क और र्ॉडल-वसम्युलेिेड 

र्लू्यों के बीच रै्च ८% के भीतर थे। यह प्रदवर्ात वकयम गयम ह ैवक र्ॉडल कम उपयोग लडएफ और यएूफ ऑपरेर्ि के बमद लक्ष्य अंवतर् एकमग्रतम 

के वलए प्रमरंवभक / विविर्य बफर एकमग्रतम की भविष्यिमणी करि ेके वलए वकयम जम सकतम ह।ै एक्ससलिएिंट्स के बहमि पर वसस्िर् र्मपदंडों कम 

प्रभमि र्ॉडल से वििमाररत वकयम गयम थम, और एक्ससलिएिंट्स और ियाबस चमजा के विचलि को सबस ेर्जबूत प्रभमि पमयम गयम थम। 

हर्मरम र्मििम ह ैवक इस थीवसस के एक भमग के रूप र्ें यहमं प्रस्तुत अध्ययि जैि-तंत्र के विविर्माण के वलए अत्यविक कुर्ल पैि उपकरणों के 

विकमस को बढमिम देगम। 
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